Author’s response to reviews

Title: Antihypertensive potential of the aqueous extract which combine leaf of Persea americana Mill. (Lauraceae), stems and leaf of Cymbopogon citratus (D.C) Stapf. (Poaceae), fruits of Citrus medical L. (Rutaceae) as well as honey in ethanol and sucrose experimental of model

Authors:

- Paul Désiré D Dzeufiet (dzeufiet@yahoo.fr)
- Amelie Moguéo (amelie.mogue@yahoo.fr)
- Danielle Claude Bilanda (bilandad@yahoo.com)
- Bibi-Farouk O Aboubakar (bibifarouck@yahoo.fr)
- Léonard Tedong (ltedong@yahoo.fr)
- Théophile Dimo (dimo59@yahoo.com)
- Pierre Kamtchouing (pikam55@yahoo.fr)

Version: 5 Date: 24 November 2014

Author’s response to reviews: see over
Dear Editor in Chief

Please find the final correction of our manuscript entitled “Antihypertensive potential of the aqueous extract which combine leaf of Persea americana Mill. (Lauraceae), stems and leaf of Cymbopogon citratus (D.C) Stapf. (Poaceae), fruits of Citrus medical L. (Rutaceae) as well as honey in ethanol and sucrose experimental of model”

Title page and author’s contribution have been added as requested and are in red in the manuscript.

Best regards.